Author’s response to reviews

Title: HIV preventive behavior and associated factors among mining workers in Sali traditional gold mining site Bench Maji zone, Southwest Ethiopia: A cross sectional study.

Authors:

Hordofa Gutema Abdissa (pthordeg@gmail.com)
Yohannes Kebede Lemu (yohanneskbd@gmail.com)
Dejene Tilahun Nigussie (dejetn@yahoo.com)

Version: 2 Date: 27 March 2014

Author’s response to reviews:

Cover letter
Hordofa Gutema
Mizan Aman General Hospital
Bench Maji Zone, Mizan Teferi
Email: pthordeg@gmail.com

To BMC Journal editorial office
March 27, 2014
Dear Jane:

I am very pleased to resubmit the research article entitled “HIV preventive behavior and associated factors among mining workers in Sali traditional gold mining site Bench Maji zone, Southwest Ethiopia: A cross sectional study.”

I am very sorry for being late to resubmit this manuscript after request for adjustment was given on line numbers, page number and removal of the sections. For the last one week there was no internet access in my working area and didn’t get your mail. Promising such delaying never to happen again, I have arranged the as the comment given.

The line numbers has given for all pages in continues manner, the page number also given for each pages, the section has been removed from each section. Looking forward to hearing from you.

Thank you again for your consideration!

Sincerely,
Hordofa Gutema (Bsc, MPH)